% Instructions:
%
% Modify this document until it looks exactly like the sample solution.
% Note that parts of the solution have been replaced with the word TODO.

\documentclass[11pt]{article}
\usepackage[margin=1in]{geometry}
\setlength{\parindent}{0pt}
\setlength{\parskip}{4pt}

\title{LaTeX Scavenger Hunt}
\author{Dr. Nathan Sprague}
\date{January 15, 2014}

\usepackage[pdftex]{hyperref}

\begin{document}

\begin{center}
{ \Large \textbf{\LaTeX} Scavenger Hunt }
\end{center}

\section*{Ex 1}

Figure out how to type a sentence with bold, italics, and underline.

\section*{Ex 2}

See if you can type TODO inline.

Then get it to display on its own line: TODO

Hint: you should NOT use \texttt{\textbackslash{}begin\{center\}}

\section*{Ex 3}

Make a bulleted list (A footnote would be nice, too.) with at least
three items:
JMU has an excellent CS department.
CS228 makes you smarter.
Logic works.

\section*{Ex 4}

See if you can get both "quotes" to point the right way.

Also, do you know the difference between TODO, TODO, and TODO?

\section*{Ex 5}

Figure out how to make a truth table. (Hint: Wizard $\rightarrow$
Quick Tabular)

TODO

\vspace{1em} P.S. You may find
\url{http://www.stdout.org/~winston/latex/} useful for hw \#1.

\end{document}
